Sat Apr 28 16:46:48 GV 2001
Year: 01 Doy: 118
Cbserver: el nore

Sat Apr 28 16:52:57 GVI 2001 CH P Start Patro
Sat Apr 28 16:52:59 GMIr 2001 PI CS Start Patro
Sat Apr 28 17:02:19 GVI 2001 CH P LSD

WEATHER COMMVENT: Sat Apr 28 17:01:37 GMIr 2001

Sone cirrus scattered about the sky. Light west w nd

50F.

**VKIV PROBLEMF*: Sat Apr 28 17:04:01 GVr 2001

KCC crashed and 'reset’ on processor board had no effect. Powered
of f chassis and it appears to work

Sat Apr 28 17:05:45 GVII 2001 CH P End LSD

Sat Apr 28 17:05:51 GVI 2001 CH P Bi asLSD

Sat Apr 28 17:06:34 GVI 2001 CH P End Bi asLSD
Sat Apr 28 17:06:41 GV 2001 CH P Bi as

Sat Apr 28 17:07:27 GJI 2001 CH P End Bi as

Sat Apr 28 17:08:34 GVI 2001 Pl CS End Patro
Sat Apr 28 17:11:47 GMI 2001 MKI VvV Start Correct
Sat Apr 28 17:11:54 GVII 2001 Pl CS Start Patro

Sat Apr 28 17:13:21 GVI 2001 MKI V End Correct

Sat Apr 28 17:13:36 GVI 2001 MKI V Start Patro

MKIV COMMENT: Sat Apr 28 17:12:02 GMI 2001

Ol gui der work by Judd Johnson is conplete.

RA nmotor gear was stripped. Plastic replaced with neta

one. Possibly as a result phase of mpbtor and cam was nessed

up. Recentered so that guiding was possible.

Limt switches installed so that notor would not attenpt to
drive past limt. This elimnated the bangi ng of the notor

into the limt that resulted in the stripped gear

The tine constant of the integrator circuit was reduced slightly.
See Judd for details.

The back end of the guide tel escope was |oose. It could rotate and
was not secure in declination. Push pull screws were installed in
declination and rotation was tightened. This should influence
gui di ng changes during the day. -- david

Sat Apr 28 17:18:59 GMI 2001 MKI VvV Start Patro

MKI'V COMMENT: Sat Apr 28 17:19:16 GVI 2001

Canera has been replaced: 12 to canera optics aligned. Mangin
mrror repositioned so that return beamis nore on axis.

Slit wi dths have been changed. Fornerly widths at |ower and
upper hairlines were 212 and 383 microns. Now wi dths are

75 and 140 microns. Gain of canera and hopefully better optica
alignment required the change. Wth new alignnent there is nore
intensity at |ow heights and at upper heights.



MKI'V COMMENT: Sat Apr 28 17:23:33 GVI 2001

Correct performed. Check values. Note the dark offsets are much

| arger than before. Also note the gain values are at |low |l evels

over much nore of the range. | still have a concern that the

Correct routine is not properly spanning the hairlines.

MKI'V COMMENT: Sat Apr 28 17:28:19 GVII 2001

Cirrus during last scan. Look for bands noving through

MKIV COMMENT: Sat Apr 28 17:28:47 GMI 2001

Scan azimuth has not been fixed. INfact it is worse, approxinmately

12 degrees off.

**VKIV PROBLEMF*: Sat Apr 28 17:29:44 GMr 2001

The resident tscp file and calibration are unable to produce a neani ngfu

real tinme display.

**ECHO PROBLEM*: Sat Apr 28 17:34:29 GMI 2001

Not pointed on sun, guiding in Ephem node. Light |evel shows about 12.1 but

it usually shows much higher when the sun is clear like this. It |ooks like
somet hi ng has degraded the guiding since Thursday maybe. On Thursday it seened
i ke the guiding would go back to Quad C nuch sooner after a cloud passed over.

Sat Apr 28 18:01:25 GVI 2001 CH P LSD

Sat Apr 28 18:03:15 GMr 2001 PI CS Fl at

Sat Apr 28 18:05:04 GMI 2001 CH P End LSD

Sat Apr 28 18:05:10 GVI 2001 CH P Bi asLSD

Sat Apr 28 18:05:58 GVI 2001 CH P End Bi asLSD
Sat Apr 28 18:06: 04 GVI 2001 CH P Bi as

Sat Apr 28 18:06:16 GVI 2001 Pl CS End Fl at

Sat Apr 28 18:06:49 GMI 2001 CH P End Bi as

**ECHO PROBLEM *: Sat Apr 28 18:14:24 GMI 2001

I Just saw the "Quad C' guiding kick in after the light |evel got above 18. 35,
the threshold may have been | ower but that was the value when | | ooked at it
and | saw the inages junp fromthe corner of the screen to the correct position,
i ndicating that the quadcell guiding had just started. So maybe the threshol d
is the same as before but it sure took a long tine for the light |evel reading
to get up there. WMaybe whatever is measuring the |light |evel sees the sun
easi er when the tel escope is off-pointed in certain directions.

Sat Apr 28 18:29:53 GVI 2001 MKI V Start Patro

Sat Apr 28 18:37:39 GVI 2001 MKI V End Correct

Sat Apr 28 18:38:09 GVMr 2001 MKI VvV Start Patro

Sat Apr 28 18:45:05 GMIr 2001 MKI 'V End Patro

Sat Apr 28 18:48:06 GMI 2001 MKI V Start Patro

MKI'V COMMENT: Sat Apr 28 18:57:55 GV 2001

Phased the FLC for maxi num nodul ati on at 240Hz.

Did a Correct.

Realigned S-K lens to maxi n ze signa

D d anot her Correct.

MKIV COMMENT: Sat Apr 28 18:58:55 GMI 2001

Calibration in cirrus just to see if we can cal. First corona

i nbet ween scans is bad, but coronal scan just before cal apperas

to be OK. It could be used as the coronal reference for the



calibration .

Sat Apr 28 19:03:14 GMI 2001 CH P LSD

Sat Apr 28 19:06:38 GMI 2001 CH P End LSD

Sat Apr 28 19:06:48 GMI 2001 CH P Bi asLSD

Sat Apr 28 19:07:40 GMI 2001 CH P End Bi asLSD
Sat Apr 28 19:07:53 GMI 2001 CH P Bi as

Sat Apr 28 19:08:47 GVI 2001 CH P End Bi as

Sat Apr 28 19:22:57 GMI 2001 MKI V Start Patro

MKIV COMMENT: Sat Apr 28 19:36:41 GMTI 2001

Pausing to correct barrel azimuth.

We can’t believe that we can see the corona through the anpunt

of cirrus in the sky!

COMMENT: Sat Apr 28 19:43:01 GvVIr 2001

Ext ended the dome sl ot.

Sat Apr 28 19:46:40 GMI 2001 Pl CS End Patro

Sat Apr 28 19:46:56 GVI 2001 CH P End Patro

COMMENT: Sat Apr 28 19:46:59 GV 2001

Spar off pointed for nk4 work, paused instrunents.

Sat Apr 28 20:36:42 GMI 2001 MKI VvV Start Patro

Sat Apr 28 20:37:37 GMI 2001 CH P Start Patro

Sat Apr 28 20:37:39 GMI 2001 PI CS Start Patro

MKI'V COMMENT: Sat Apr 28 20:37:45 GVII 2001

Done wi th sonme adjustnents, doing a height mask test now. Spar is pointed
and tracking again, restarted the instrunents.

MKI'V COMMENT: Sat Apr 28 20:41:23 GV 2001

Hei ght Mask Test: Opal gives a healthy signal for all three

sl ots and above the top of the mask.

Sat Apr 28 20:51:26 GMI 2001 MKI VvV Start Patro

MKI'V COMMENT: Sat Apr 28 20:50: 33 GV 2001

Corrected azinmuth reading so that it now reads 12 degrees |less than
before. This is the 9.5 degree error from conpari sons w th other
instruments plus the 2.5 degree error introduced yesterday.

MKI V COMMENT: Sat Apr 28 20:52:35 GMI 2001

Sky | ooks as good as we have seen. Intensity trace on scope does not
| ook great. This inmplies the lens is not clean

Sat Apr 28 21:03:14 GVI 2001 MKI V Start Patro

Sat Apr 28 21:04:22 GV 2001 CH P LSD

MKIV COMMENT: Sat Apr 28 21:06:19 GMI 2001

We |ike the change in azinuth conpared to LASCO C2.

MKI'V COMMENT: Sat Apr 28 21:06: 37 GV 2001

Previous calibration fromtoday no | onger applies. Q and U gains
have been increased by 4X in software.

001

Sat Apr 28 21:07:46 GVI 2001 CH P End LSD
MKIV COMMENT: Sat Apr 28 21:07:44 GV 2001

I will soon clean the QL.

Sat Apr 28 21:07:56 GVI 2001 CH P Bi asLSD

Sat Apr 28 21:08:44 GVI' 2001 CH P End Bi asLSD



Sat Apr 28 21:
Sat Apr 28 21:
Sat Apr 28 21:

MKl V COVIVENT:
Cl eaned front

Sat Apr 28 21:
Sat Apr 28 21:

MKI'V COMVENT:

correct for the bandpass we get
COMMVENT: Sat Apr

08: 53 GvIr 2001
09: 39 GV 2001
48:16 GMI 2001
Sat Apr
surface of Ol.
55:56 GMI 2001
56: 25 GMI 2001
Sat Apr

CH P
CH P
MKI V

Bi as
End Bi as
Start Patrol

28 21:48:28 GMI 2001
Scanning to sOL is now clean and installed, Spar is guiding.

CH P
PI CS

End Patr ol
End Patr ol

28 23:08:11 GV 2001
Did some Hartman mask tests and found that the OL/Ccculter distance is currently

Darryl is heading down for the day.

Sat Apr 28 23:12:09 GMI 2001

M1V

17 _19.rawnk4
17 _22.rawnk4
17_25. rawnk4
17_28. rawnk4
17_31. rawrk4
17_34. r awrk4

18_29.
18_32.
18_41.
18_48.
18 51.
18_57.

r awk4
r awk4
r awk4
r awk4
r awnk4
r awnk4

19_03.
19_10.
19_22.
19_26.
19_29.
19_31.

r awk4
r awk4
r awk4
r awrk4
r awnk4
r awnk4

in our data, yay!
28 23:11:30 GV 2001

20_36.
20_42.
20 _51.
20 _54.
21 03.
21_06.

r awk4
r awk4
r awk4
r awrk4
r awnk4
r awnk4

21 48. rawk4
21 51. rawk4
c18 54.rawnk4
c19 00.rawnk4
cl19 07.rawnk4



